Author's response to reviews

Title: Adenoma of Vater Ampulla. A case report study

Authors:

Konstantinos Alexiou (knegro@otenet.gr)
Argyrios Ioannidis (argioannid@med.uoa.gr)
Ioannis Drikos (johndrikos@yahoo.com)
Athanasios Fotopoulos (athfot@gmail.com)
Ioannis Karanikas (ikaranikas@yahoo.gr)
Nikolaos Economou (gramdiy@sismanoglio.gr)

Version: 6 Date: 20 March 2014

Author's response to reviews:

After reading through your manuscript, we feel that the quality of written English needs to be improved before the manuscript can be considered further.

Done